Author's response to reviews

Title: Prognostic factors of successful tympanoplasty in pediatric patients: cohort study

Authors:

Nuria Esperanza Boronat-Echeverría (boenorl@live.com.mx)
Esmeralda Reyes-García (esme.schmoozykatzen@gmail.com)
Yolanda Sevilla-Delgado (yosevilla@hotmail.com)
Hector Aguirre-Mariscal (Hectororl@hotmail.com)
Juan Manuel Mejía-Arangure (juan.mejiaa@imss.gob.mx)

Version: 2 Date: 12 March 2012

Author's response to reviews:

Rachel Neiman
BMC Pediatrics
Dear Dr. Neiman,

We submit our manuscript, entitled “Prognostic factors of successful tympanoplasty in pediatric patients: Cohort study”, for consideration for publication in BMC Pediatrics by Boronat Echeverría et al.

We review our manuscript and add the next statement in methods section: Written informed consent for participation in the study was obtained from participants' parents.

After of conclusion we add: Competing interests: The authors declare that they have no competing interests

In abstract we add in background: The success of tympanosplasty traditionally has been measured only by the post-operative integrity of the graft. Yet, there exist other variables to consider the success. We include a methods section.

We thank you in advance for your time in considering this article for publication in BMC Pediatrics

Sincerely,

Dr. Juan Manuel Mejía-Aranguré
Senior Researcher
Clinical Epidemiology Research Unit, High Specialty Medical Unit,
Pediatric Hospital, National Medical Center "Century XXI"
National Commission of Scientific Research of the Mexican Institute of Social
Security